Author's response to reviews

Title: The involvement of men in maternal health care: participatory case studies from Maligita and Kibibi, Uganda

Authors:

Debra Singh (ahderom@gmail.com)
May Lample (may_mag9@gmail.com)
Jaya Earnest (J.Earnest@curtin.edu.au)

Version: 2
Date: 20 January 2014

Author's response to reviews: see over
Dear Sir/Madam,

We are submitting a paper entitled: The involvement of men in maternal health care: participatory case studies from Maligita and Kibibi, Uganda, based on focus group and questionnaires that we undertook at the beginning of 2013. The issue of male involvement in maternal health issues remains critical for improved maternal health outcomes in countries like Uganda. As a relatively small local NGO based in Uganda we are not however in a position to pay for the fee should the paper be accepted. We therefore kindly request a waiver.

We greatly appreciate your considering our submission.

Yours sincerely,

Dr. Debbie Singh
BSc, Grad Dip Diet, BMBS, MPH, RACGP